Craig Chambers CSE 505 (Fall ’95) Homework Assignments

Homework Assignment #6

Due Monday 11/20

1.Write filter_stream in both Scheme and ML, extending the stream data structures defined in class.

2.Implement a stream of the fibonacci numbers, in either Scheme or ML. Use an algorithm where the cost of computing each fibonacci number in the sequence is constant.

3.In ML, implement a binary tree that takes a function to do comparisons. Support an interface where the comparison function is passed only when the binary tree is created, not at each lookup and insert operation call.

4.Implement an interpreter for the denotational semantics of the language given in Chapter 5, under lazy evaluation rules. You may choose either Scheme or ML for your implementation, but if you know one of the languages already, you should choose the language you know less well.

5.Evaluate both Scheme and ML in terms of their adherence to the Abstraction Principle and the Qualification Principle.